Author's response to reviews

Title: Psychometric evaluation of the Problem Areas in Diabetes (PAID) survey in Southern, rural African American women with Type 2 diabetes

Authors:

Stephania T Miller (smiller@mmc.edu)
Tom Elasy (tom.elasy@vanderbilt.edu)

Version: 3 Date: 15 February 2008

Author's response to reviews: see over
February 15, 2008

Dear Editor:

Please find attached our revised manuscript with the title “Psychometric evaluation of the Problem Areas in Diabetes (PAID) survey in Southern, rural African American women with Type 2 diabetes”, MS: 1112032722156788.

Per your guidance, we have 1) reviewed the title to insure that it is in sentence case, 2) removed the authors' qualifications from the manuscript and submissions system, 3) included the country in the author affiliation details, 4) placed each abstract section in a separate paragraph, 5) included the abstract “purpose” in the abstract background section, and 6) included a Discussions section between the Results and the Conclusions.

My co-author and I would like to thank you for the opportunity to publish our work and we look forward to hearing from you soon regarding final acceptance.

Sincerely,

Stephania T. Miller
Tom A. Elasy